X-Git-Url: https://bilbo.iut-bm.univ-fcomte.fr/and/gitweb/16dcc.git/blobdiff_plain/d69591c41135e899d27072006db6af016df62445..79f46c777f6ef3202c2fb25194822518cad3e32c:/preliminaries.tex?ds=sidebyside

diff --git a/preliminaries.tex b/preliminaries.tex
index 1200bc5..35d48a1 100644
--- a/preliminaries.tex
+++ b/preliminaries.tex
@@ -57,7 +57,7 @@ Figure~\ref{fig:iteration:f*}.
 
 \begin{figure}[ht]
 \begin{center}
-\includegraphics[scale=0.5]{images/iter_f0c}
+\includegraphics[width=0.45\textwidth]{images/iter_f0c}
 \end{center}
 \caption{Iteration Graph $\Gamma(f^*)$ of the function $f^*$}\label{fig:iteration:f*}
 \end{figure}
@@ -94,8 +94,16 @@ return $x$\;
 
 
 
-With all this material, we can study the chaos properties of these 
-function.
-This is the aims of the next section. 
 
+Based on this setup,
+we can study the chaos properties of these 
+functions.
+This is the aim of the next section. 
 
+
+%%% Local Variables:
+%%% mode: latex
+%%% TeX-master: "main"
+%%% ispell-dictionary: "american"
+%%% mode: flyspell
+%%% End:
